Author’s response to reviews

Title: Distal and Lateral Subungual Onychomycosis of the Finger Nail in a Neonate: a Rare Case

Authors:

Supram Hosuru Subramanya (supram.gowda@gmail.com)
Saujan Subedi (saujan.subedi2015@gmail.com)
Yang Metok (yangmetok@gmail.com)
Ajay Kumar (drajayk2002@yahoo.com)
Peralam Prakash (prakashpy123@yahoo.co.in)
Niranjan Nayak (niruni2000@yahoo.com)

Version: 4 Date: 17 May 2019

Author’s response to reviews:

1. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Authors response: changes are made as suggested.

2. Thank you for providing a CARE document. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

Authors response: CARE document is removed from submission.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors response: changes are made as suggested.